

## **Author's response to reviews**

**Title:** Multiple imputation for estimation of an occurrence rate in cohorts with attrition and discrete follow-up time points: a simulation study

### **Authors:**

Noémie Soullier ([noemie.soullier@inserm.fr](mailto:noemie.soullier@inserm.fr))  
Elise de La Rochebrochard ([roche@ined.fr](mailto:roche@ined.fr))  
Jean Bouyer ([jean.bouyer@inserm.fr](mailto:jean.bouyer@inserm.fr))

**Version:** 5 **Date:** 2 September 2010

### **Author's response to reviews:**

Changes have been made according to your mail sent 2 September 2010